Author's response to reviews

Title: Intracellular expression of toll-like receptor 4 in neuroblastoma cells and their unresponsiveness to lipopolysaccharide

Authors:

Ferdaus Hassan (hassan@aichi-med-u.ac.jp)
Shamima Islam (shamima@aichi-med-u.ac.jp)
Gantsetseg Tumurkhuu (gtsetseg@aichi-med-u.ac.jp)
Yoshikazu Naiki (naikiy@aichi-med-u.ac.jp)
Naoki Koide (nao777nk@aichi-med-u.ac.jp)
Isamu Mori (isamor@aichi-med-u.ac.jp)
Tomoaki Yoshida (tomo@aichi-med-u.ac.jp)
Takashi Yokochi (yokochi@aichi-med-u.ac.jp)

Version: Date: 18 July 2006

Author's response to reviews:

Dear the Editor:

Please find attached the manuscript entitled "Intracellular expression of toll-like receptor 4 in neuroblastoma cells and their unresponsiveness to lipopolysaccharide" written by Ferdaus Hassan et al. The manuscript deals with expression and localization of TLR4 in human neuroblastoma cell line NB-1 and its unresponsiveness to LPS. To our knowledge, this is the first report of presence of TLR4 in neuroblastoma cell. I hope that our paper is suitable for BMC Cancer. Thank you for handling our manuscript.

Yours sincerely,

Re: BMC Cancer

Takashi Yokochi, M.D., Ph.D.
Department of Microbiology and Immunology
Aichi Medical University, Nagakute,
Aichi 480-1195, Japan
Phone: +81-561-62-3311, ext. 2269.
Fax: +81-561-63-9187.
E-mail: yokochi@aichi-med-u.ac.jp